Author's response to reviews

Title: Unusual cause of a painful right testicle in a 16-year-old male: a case report

Authors:

Khalid N Shehzad (khalidshehzad@hotmail.com)
Amjid A Riaz (mrariaz@hotmail.com)

Version: 4 Date: 20 October 2010

Author's response to reviews:

Dear Sir/Madam,

Please find attached the revised version of our manuscript entitled 'Unusual cause of a painful right testicle in a 16-year-old male - a case report' in light of the changes that you recommended.

We have made the following changes:

1) We have added authors' e-mail and postal addresses and a statement regarding affiliations on the title page.
2) The patient's ethnicity has been included in the main text as well as abstract.
3) The abstract has been included in the main text.
4) The figure legend has been moved to after the references.
5) Section headings (Introduction, Case presentation) have been added to the main text.
6) Statements regarding consent, competing interests and authors' contributions have been included.

Thank you and kind regards,

Khalid N Shehzad
Amjid A Riaz